Author's response to reviews

Title: Abdominal obesity and low physical activity are associated with insulin resistance in overweight adolescents: A cross-sectional study

Authors:

Claudia M Velasquez R (claudia.velasquez@udea.edu.co)
Marcela Velasquez V (marce.villa21@gmail.com)
Leidy J Gomez O (leidygomez0217@gmail.com)
Juliana A Bermudez C (andrea.ber879@gmail.com)

Version: 3
Date: 26 August 2014

Author's response to reviews: see over
Medellin (Colombia), August 26th 2013

Editor
BMC Pediatrics

ISSUE. Response to referee 2, MS: 2363688961302435 Abdominal obesity and low physical activity are associated with insulin resistance in overweight adolescents: A cross-sectional study

Evaluator N° 2: Loreana Silveira

I have recommended before, as a major compulsory revision, In table 1, the necessity of correction in sex percentage of NW group to 47.5. In the cover letter, the authors agreed with my revision, but in the manuscript uploaded, this table remains unchanged.

Response to the reviewer: Table 1 was corrected. Male: 52.5% female: 47.5%.

Sincerely yours

Claudia Maria Velásquez R.
Professor Universidad de Antioquia
Medellín, Colombia